Author’s response to reviews

Title: High-viscosity bone cement for vertebral compression fractures: a prospective study on intravertebral diffusion and leakage of bone cement

Authors:
meiyong wang (wanglei10008@sina.com)
qunhua jin (11083861@qq.com)

Version: 2 Date: 24 Jul 2020

Author’s response to reviews:
1. Declarations section have been added.
2. The original data has been added in the 'Availability of data and materials' section.
3. I have submit a clean version.
4. Empty pages have been removed.